Author’s response to reviews

Title: EQUIP: Implementing chronic care principles and applying formative evaluation methods to improve care for schizophrenia

Authors:

Alison H Brown (alisonh@ucla.edu)
Amy N Cohen (ancohen@ucla.edu)
Matthew J Chinman (chinman@rand.org)
Christopher Kessler (Christopher.Kessler@va.org)
Alexander S Young (ayoung@ucla.edu)

Version: 4 Date: 9 September 2007

Author’s response to reviews: see over
Response to Reviewer Martin Eccles

Minor Essential Revision

The term “opinion leader” has now been changed to “product champion” in the abstract and in several places in the manuscript.